Title: Mandibular facial talon cusp: Case report.

Authors:

Folakemi A Oredugba (tored@nova.net.ng)

Version: 6 Date: 11 November 2005

Author's response to reviews: see over
November 09 2005

The Editorial Team

BioMedCentral.

Dear Sir,

**Re: Mandibular facial talon cusp: case report**

( MS: 1022510845699242)

I received your mail dated November 04 2005 on the re-review of the above named manuscript. I have made the following changes as suggested:

1. **Title:** The title has been changed to "**Mandibular facial talon cusp: case report**".

2. **Title Page:** Has been corrected to appear as in the instructions for authors.

3. **Title and Abstract:** These are now identical in the manuscript and in the submission system.

4. **Legends to Figures** has been changed to **Figures** as in the formatting checklist and legends typed under. The same applied to "**Table**".

Thank you.

Sincerely,

Dr Folakemi A Oredugba